Author’s response to reviews

Title: Inhibitory effects of ChondroT and its constituent herbs on RANKL-induced osteoclastogenesis

Authors:

Rui Hong Guo (guorhyx@naver.com)
Seon-Jong Kim (mofoster@hanmail.net)
Chan-hun Choi (mensolog@hanmail.net)
Chang-su Na (nakugi@hanmail.net)
Bok Yun Kang (bykang@chonnam.ac.kr)
Young Ran Kim (kimyr@chonnam.ac.kr)

Version: 4 Date: 30 Oct 2019

Author’s response to reviews:

Editor Comments:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


https://www.spandidos-publications.com/mmr/13/5/4014?text=fulltext

This overlap exists throughout your manuscript.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

Answer: Thank you for your comments. We are very sorry about the textual overlap with other previously published papers. We have corrected them and confirmed the overlap using Turnitinn on the school library website. We revised some sentence expressions without changing our idea.
As to the reference 2 you pointed out (https://www.frontiersin.org/articles/10.3389/fphar.2018.01441/full), it was published by our lab and written by the same person, possibly using some of the same materials and the same methods. We have tried our best to revise the manuscript. However, there are still some abbreviations of the terminology and their full names in this manuscript, and the experimental materials used cannot be changed, so there may also have some overlap with other papers.

2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Answer: Thank you for your comments. We have added this content in the funding section as “The funding body did not participate in the design of this study and collection, analysis and interpretation of the data as well as the writing of the manuscript”.

3. Please put your responses to the reviewers’/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer: Thank you for your comments. We have uploaded our manuscript according to your indications.